Author's response to reviews

Title: Genetic polymorphisms in folate pathway enzymes, DRD4 and GSTM1 are related to temporomandibular disorder

Authors:

Angel Aneiros-Guerrero (angelaneiros@hotmail.com)
Ana M Lendinez (analendinezramirez@gmail.com)
Arturo R Palomares (arturolrp@gmail.com)
Beatriz Perez-Nevot (bpnevot@gmail.com)
Lidia Aguado (somoscromosomas@hotmail.com)
Alvaro Mayor-Olea (dralvaromayor@gmail.com)
Maximiliano Ruiz-Galdon (mruizg@uma.es)
Armando Reyes-Engel (engel@uma.es)

Version: 4 Date: 12 May 2011

Author's response to reviews: see over
Changes made to manuscript MS: 8294055034719755

Major revisions

1. Author list: the list of authors in the manuscript has been written as they are in the submission system (First name Initial Last name).

2. Authors' contributions: We have indicated that all authors read and approved the final manuscript. This section has been formatted into a paragraph rather than a list.

3. References: References have been cited in the text using consecutive numbers in square brackets. Superscript numbers have been suppressed (it is related to reference 28 on page 6). References are cited in numerical order. Specifically, references 32 and 44 have been cited in the main text that were not in the previous manuscript.

4. Additional files have been removed and figure 1, tables 1, 2 and 3 are included in the manuscript.

Minor revisions

a. Highlighting: All highlighting has been removed from the manuscript.

b. References: The term 'et al.' has been removed from reference 11.

c. Figures: The figure has been removes from the manuscript file (just leaving the title and the legend).

d. Typography: We have checked the manuscript for typographical errors and made final corrections.